Author's response to reviews

Title: Occupational therapy and return to work: a systematic literature review

Authors:

Huget A M Désiron (Huget@act-desiron.be)
Angelique de Rijk (angelique.derijk@maastrichtuniversity.nl)
Elke Van Hoof (Elke.VanHoof@wiv-isp.be)
Peter Donceel (peter.donceel@med.kuleuven.be)

Version: 3 Date: 27 July 2011

Author's response to reviews:

Dear Editor,

We thank you for accepting our manuscript. According to your instructions, we have made the following changes:

1. Major revisions

   The title and abstract are now identical in the manuscript and the submission system.

   The list of authors in the manuscript is now written exactly as they are in the submission system, both in style and order.

   Figure title was included.

   Titles were provided for all tables.

   We have removed the additional files and the tables were included in the manuscript in a tables section following the references.

2. Minor revisions

   Highlighting/tracking were removed from the manuscript.

   Comments were removed.

   The figure was adapted.

Yours sincerely,

Peter Donceel and Huget Desiron